Author's response to reviews

Title: A national survey of the prevalence of schistosomiasis and soil transmitted helminths in Malawi

Authors:

Cameron Bowie (cam.bowie@malawi.net)
Bernadette Purcell (purceb@doctors.org.uk)
Bina Shaba (bilharziacontrol@malawi.net)
Peter Makaula (chisale@malawi.net)
Maria Perez (arantza_perez@yahoo.com)

Version: 5 Date: 11 November 2004

Author’s response to reviews: see over
All format changes have been made as requested